Author's response to reviews

Title: The challenge of comprehensively mapping children's health in a nation-wide health survey: design of the German KiGGS-Study.

Authors:

Baerbel-Maria Kurth (KurthB@rki.de)
Panagiotis Kamtsiuris (KamtsiurisP@rki.de)
Heike Hoelling (HoellingH@rki.de)
Martin Schlaud (SchlaudM@rki.de)
Ruediger Doelle (DoelleR@rki.de)
Ute Ellert (EllertU@rki.de)
Heidrun Kahl (KahlH@rki.de)
Hiltraud Knopf (KnopfH@rki.de)
Michael Lange (LangeM@rki.de)
Gert BM Mensink (MensinkG@rki.de)
Hannelore Neuhauser (NeuhauserH@rki.de)
Angelika Schaffrath-Rosario (Schaffrath-RosarioA@rki.de)
Christa Scheidt-Nave (Scheidt-NaveC@rki.de)
Liane Schenk (liane.schenk@charite.de)
Robert Schlack (SchlackR@rki.de)
Heribert Stolzenberg (StolzenbergH@rki.de)
Michael Thamm (ThammM@rki.de)
Wolf Thierfelder (ThierfelderW@rki.de)
Ute Wolf (WolfU@rki.de)

Version: 3 Date: 21 May 2008

Author's response to reviews: see over
Dear editors,

We hope that all your hints and questions are considered now:

A title page with all the information you wanted is included, we removed figure and replacement texts in our manuscript as well as the titles in our figures and tables. The additional files are mentioned in the text. Few typing and content errors are corrected now.

Sincerely yours

[Signature]

Head of the department of health reporting and epidemiology
Project leader of the KiGGS study